Author's response to reviews

Title: Measuring nasal bacterial load in children at low and high risk of tympanic membrane perforation

Authors:

Heidi C Smith-Vaughan (heidi@menzies.edu.au)
Roy Byun (roybyun@dental.wsaahs.nsw.gov.au)
Mangala Nadkarni (mnadkarni@dental.wsaahs.nsw.gov.au)
Nicholas A Jacques (njacques@dental.wsaahs.nsw.gov.au)
Neil Hunter (nhunter@dental.wsaahs.nsw.gov.au)
Stephen Halpin (stephen.halpin@menzies.edu.au)
Peter S Morris (peter.morris@menzies.edu.au)
Amanda J Leach (amanda@menzies.edu.au)

Version: 2 Date: 6 November 2005

Author's response to reviews:

Dear Sir/Madam

Our manuscript entitled, "Measuring nasal bacterial load in children at low and high risk of tympanic membrane perforation", was previously submitted to BMC Microbiology. However, we were advised that it was more appropriate for BMC Ear, Nose and Throat Disorders.

Please accept this manuscript for consideration for publication in your journal. The only modification of the previous version was in providing the requested format for BMC Ear, Nose and Throat Disorders.

Best wishes
Heidi Smith-Vaughan